Mauna Loa Solar Observatory Observer's Log _____ Tue Dec 26 17:24:21 GMT 2017 Year: 17 Doy: 360 Observer: rose WEATHER COMMENT: rose: Tue Dec 26 17:49:15 GMT 2017 Temp: 40.2f, Humidity: 19%, Pressure: 28.726in, Wind: 6mph from 175degs, Skies: clear end GENERAL COMMENT BY rose: Tue Dec 26 17:49:41 GMT 2017 PM Blew off Kcor O1 end GENERAL COMMENT BY rose: Tue Dec 26 17:49:55 GMT 2017 PM Blew off CoMP 01 end **GENERAL PROBLEM COMMENT BY rose** : Tue Dec 26 17:54:57 GMT 2017 MLSO comment Logger was not working again this morning. It would not allow me to log comments. It is working now. end Tue Dec 26 17:56:19 GMT 2017 Kcor Focus/alignment program exited Tue Dec 26 18:00:31 GMT 2017 KCOR Start Synoptic Patrol Tue Dec 26 18:01:04 GMT 2017 CoMP occulter has been re-centered Tue Dec 26 18:01:10 GMT 2017 COMP Start Patrol on cookbook: synoptic-00012.cbk Tue Dec 26 19:08:42 GMT 2017 KCOR End Patrol Tue Dec 26 19:08:44 GMT 2017 KCOR Start Calibration script: c:\kcor\mlso-calibration22deg-20171025.ini Tue Dec 26 19:24:02 GMT 2017 KCOR End Calibration Script Tue Dec 26 19:24:19 GMT 2017 KCOR Start Synoptic Patrol Tue Dec 26 19:24:20 GMT 2017 KCOR Start Synoptic Patrol Tue Dec 26 19:35:47 GMT 2017 COMP End Patrol Tue Dec 26 19:35:51 GMT 2017 COMP Start Patrol on cookbook: synoptic-00012.cbk GENERAL COMMENT BY rose: Tue Dec 26 20:33:06 GMT 2017 cirrus clouds moving in coverage 50% now end Tue Dec 26 20:56:21 GMT 2017 KCOR End Patrol Tue Dec 26 20:56:41 GMT 2017 CoMP Paused for clouds GENERAL COMMENT BY rose: Tue Dec 26 20:57:56 GMT 2017 pausing for cirrostratus pushing in from the south end Tue Dec 26 23:38:15 GMT 2017 CoMP Restarted from pause Tue Dec 26 23:38:19 GMT 2017 COMP End Patrol COMP COMMENT BY rose: Wed Dec 27 00:14:36 GMT 2017 COMP VPN was connected so no data transfered to Boulder. It is now disconnected. end ONSITE STAFF: rose